\begin{table}%T7
\scriptsize
\caption{\label{tab:phot} Optical photometry of objects in the \object{Cha~II} dark cloud.}
\begin{tabular}{lccccccccccl}
%\setlength{\tabcolsep}{5pt}
%\setlength{\tabcolsep}{2pt}
\hline\hline
Designation & RA J2000        & Dec J2000       & $R_{\rm c}$ & $I_{\rm c}$ & $z$ & $\rm H\alpha_{7}$ & H$\alpha_{12}$ & $m_{856}$ & $m_{914}$ & Main Ref. & Note \\ 
            & (hh:mm:ss)     & (dd:mm:ss)     &	      &	            &     & 	          & 	           &	       &	   &           &            \\ 
\par
\hline
\object{IRAS~12416-7703}		& 12:45:06.43 & $-$77:20:13.52 &     S  	      &        S	     &    S	      & 11.86~$\pm$~0.09 & 12.10~$\pm$~0.13 &   S		 &   S  	  & i	& $\bullet$			      \\
\object{IRAS~12448-7650}		& 12:48:25.70 & $-$77:06:36.72 &     S  	      &        S	     &    S	      & 13.15~$\pm$~0.09 & 13.55~$\pm$~0.13 &   S		 &   S  	  & i	& $\bullet$			      \\
\object{IRAS~F12488-7658}		& 12:52:30.49 & $-$77:15:12.92 & 14.84~$\pm$~0.01	      & 12.21~$\pm$~0.02$^\ast$&    S	      & 15.48~$\pm$~0.10 & 15.60~$\pm$~0.13 & 12.27~$\pm$~0.09 &   S	          & f, i & $\bullet$ 			  \\
\object{IRAS~12496-7650}		& 12:53:17.17 & $-$77:07:10.63 & 16.43~$\pm$~0.01	      & 13.92~$\pm$~0.01       & 12.82~$\pm$~0.02 & 16.12~$\pm$~0.10 & 16.84~$\pm$~0.14 & 13.52~$\pm$~0.10 & 13.41~$\pm$~0.08 & a   & $\bigstar$ $\leftthumbsup$  	  \\
\object{WFI~J12533662-7706393}		& 12:53:36.62 & $-$77:06:39.31 & 22.22~$\pm$~0.14       & 19.35~$\pm$~0.04       & 18.00~$\pm$~0.04 &         ND     & 	  ND    & 19.26~$\pm$~0.20 & 18.58~$\pm$~0.12 & m   & $\leftthumbsup$$^{NEW}$  	     \\   
\object{C~17}				& 12:53:38.84 & $-$77:15:53.21 & 17.08~$\pm$~0.01       & 15.09~$\pm$~0.01       & 14.23~$\pm$~0.01 & 17.67~$\pm$~0.10 & 17.36~$\pm$~0.13 & 15.10~$\pm$~0.09 & 14.79~$\pm$~0.08 & f   & $\bullet$ $\leftthumbsup$	     \\  
\object{IRAS~12500-7658}		& 12:53:42.79 & $-$77:15:11.59 & 21.54~$\pm$~0.08       & 19.40~$\pm$~0.01       & 18.44~$\pm$~0.06 & 20.81~$\pm$~0.38 & 21.24~$\pm$~0.46 & 19.43~$\pm$~0.23 & 19.06~$\pm$~0.15 & i   & $\bigstar$ $\ding{74}$		     \\
\object{ChaII~304}			& 12:55:16.00 & $-$76:46:21.83 & 21.85~$\pm$~0.09       & 19.28~$\pm$~0.02       & 18.27~$\pm$~0.05 &       ND       &	 ND	& 19.26~$\pm$~0.17 & 18.65~$\pm$~0.11 & j	& $\bullet$ $\ding{68}$ $\leftthumbsdown$ \\
\object{ChaII~305}			& 12:55:16.48 & $-$76:46:20.89 & 21.82~$\pm$~0.09       & 19.30~$\pm$~0.01       & 18.29~$\pm$~0.05 &       ND       &	 ND	& 19.20~$\pm$~0.17 & 18.68~$\pm$~0.12 & j	& $\bullet$ $\ding{68}$ $\leftthumbsdown$ \\
\object{C~33}				& 12:55:25.72 & $-$77:00:46.62 & 16.86~$\pm$~0.01	      & 15.16~$\pm$~0.01       & 14.66~$\pm$~0.01 & 17.15~$\pm$~0.22 & 16.93~$\pm$~0.13 & 15.22~$\pm$~0.09 & 14.79~$\pm$~0.08 & f   & $\bullet$ $\leftthumbsup$  	   \\
\object{2MASS12560549-7654106}		& 12:56:05.43 & $-$76:54:10.69 & 15.95~$\pm$~0.01       & 14.54~$\pm$~0.01       & 14.04~$\pm$~0.02 & 16.17~$\pm$~0.10 & 16.15~$\pm$~0.13 & 14.58~$\pm$~0.09 & 14.42~$\pm$~0.08 & i	& $\bullet$ $\leftthumbsdown$		 \\
\object{Sz~46NW}			& 12:56:32.85 & $-$76:45:44.78 & 18.45~$\pm$~0.01	      & 17.69~$\pm$~0.17       & 17.56~$\pm$~0.04 & 18.72~$\pm$~0.13 & 18.70~$\pm$~0.14 & 17.96~$\pm$~0.10 & 18.07~$\pm$~0.09 & a	& $\bullet$ $\ding{68}$ $\leftthumbsdown$ \\
\object{Sz~46N}				& 12:56:33.59 & $-$76:45:45.18 & 14.61~$\pm$~0.01	      & 13.16~$\pm$~0.01       & 12.68~$\pm$~0.01 & 14.80~$\pm$~0.10 & 15.07~$\pm$~0.13 & 13.24~$\pm$~0.09 & 13.08~$\pm$~0.08 & a	& $\bigstar$ $\ding{68}$ $\leftthumbsup$  \\
\object{Sz~46S}				& 12:56:33.64 & $-$76:45:49.54 & 16.81~$\pm$~0.01	      & 16.05~$\pm$~0.02       & 15.93~$\pm$~0.02 & 17.07~$\pm$~0.10 & 17.03~$\pm$~0.13 & 16.32~$\pm$~0.09 & 16.38~$\pm$~0.08 & a	& $\bullet$ $\ding{68}$ $\leftthumbsdown$ \\
\object{Sz~47}				& 12:56:58.63 & $-$76:47:06.72 & 13.86~$\pm$~0.01$^\ast$& 13.38~$\pm$~0.01$^\ast$& 13.45~$\pm$~0.01 & 13.68~$\pm$~0.10 & 14.07~$\pm$~0.13 & 13.72~$\pm$~0.09 & 13.86~$\pm$~0.08 & a   & $\bigstar$ $\ding{80}$ 	 \\
\object{IRAS~12533-7632}		& 12:57:00.49 & $-$76:48:35.10 & 21.19~$\pm$~0.12       & 19.76~$\pm$~0.01       &       ND       &       ND       & 20.90~$\pm$~0.38 & 19.92~$\pm$~0.33 & 19.83~$\pm$~0.29 & e	& $\bullet$			      \\
\object{IRAS~12535-7623}		& 12:57:11.64 & $-$76:40:11.14 & 13.61~$\pm$~0.01       & 12.13~$\pm$~0.01       & 11.53~$\pm$~0.02 & 13.69~$\pm$~0.10 & 13.73~$\pm$~0.13 & 12.22~$\pm$~0.09 & 12.03~$\pm$~0.08 & e, i & $\bullet$ $\leftthumbsup$	      \\
\object{ISO-CHA~II~13}           	& 12:58:06.67 & $-$77:09:09.22 & 22.51~$\pm$~0.27	      & 19.38~$\pm$~0.01       & 17.89~$\pm$~0.04 &  ND	       &	ND	& 19.44~$\pm$~0.19 & 18.33~$\pm$~0.11 & g, k & $\bigstar$ $\leftthumbsup$		 \\
\object{WFI~J12583675-7704065}		& 12:58:36.75 & $-$77:04:06.53 & 22.45~$\pm$~0.25       & 19.32~$\pm$~0.01       & 17.85~$\pm$~0.04 &       ND       &        ND      & 19.51~$\pm$~0.20 & 18.27~$\pm$~0.10 & m   & $\leftthumbsup$$^{NEW}$  	     \\
\object{WFI~J12585611-7630105}		& 12:58:56.11 & $-$76:30:10.48 & 16.39~$\pm$~0.01       & 14.38~$\pm$~0.01       & 13.54~$\pm$~0.02 & 16.08~$\pm$~0.09 & 16.60~$\pm$~0.13 & 14.34~$\pm$~0.09 & 13.81~$\pm$~0.08 & m   & $\leftthumbsup$$^{NEW}$  	     \\
\object{C~41}                    	& 12:59:09.86 & $-$76:51:03.49 & 18.51~$\pm$~0.01	      & 17.03~$\pm$~0.01       & 16.45~$\pm$~0.02 & 18.06~$\pm$~0.10 & 18.88~$\pm$~0.14 & 17.31~$\pm$~0.09 & 17.07~$\pm$~0.08 & f,h & $\bigstar$ $\ding{80}$  	 \\
\object{ISO-CHA~II~29}		& 12:59:10.19 & $-$77:12:13.72 & 16.19~$\pm$~0.02       & 14.23~$\pm$~0.01       & 13.36~$\pm$~0.01 & 16.41~$\pm$~0.19 & 16.32~$\pm$~0.13 & 14.37~$\pm$~0.09 & 13.91~$\pm$~0.08 & g,i & $\bullet$ $\leftthumbsup$	     \\
\object{WFI~J12591881-7704419}		& 12:59:18.81 & $-$77:04:41.92 & 15.69~$\pm$~0.01       & 14.09~$\pm$~0.01       & 13.34~$\pm$~0.01 & 15.86~$\pm$~0.10 & 15.88~$\pm$~0.17 & 14.19~$\pm$~0.09 & 13.86~$\pm$~0.09 & m   & $\leftthumbsup$$^{NEW}$  	    \\
\object{WFI~J12592348-7726589}		& 12:59:23.48 & $-$77:26:58.96 & 14.41~$\pm$~0.01       & 12.99~$\pm$~0.01       & 12.39~$\pm$~0.01 & 14.78~$\pm$~0.12 & 14.62~$\pm$~0.12 & 13.13~$\pm$~0.09 & 12.75~$\pm$~0.08 & m   & $\leftthumbsup$$^{NEW}$  	    \\
\object{IRAS~12556-7731}		& 12:59:26.50 & $-$77:47:08.70 &     S		      &  S		     &	    S	      & 12.34~$\pm$~0.09 & 12.66~$\pm$~0.13 &   S		 &   S		  & i   & $\bullet$			     \\
\object{WFI~J13005297-7709478}		& 13:00:52.97 & $-$77:09:47.77 & 22.63~$\pm$~0.30       & 19.64~$\pm$~0.07       & 18.25~$\pm$~0.05 &         ND     & 	  ND    & 19.85~$\pm$~0.25 & 18.66~$\pm$~0.12 & m   & $\leftthumbsup$$^{NEW}$  	     \\ 
\object{Sz~48NE}			& 13:00:53.15 & $-$77:09:09.18 & 16.08~$\pm$~0.01	      & 14.32~$\pm$~0.04       & 13.55~$\pm$~0.01 & 16.11~$\pm$~0.09 & 16.24~$\pm$~0.13 & 14.44~$\pm$~0.09 & 14.02~$\pm$~0.08 & a	& $\bigstar$ $\ding{68}$ $\leftthumbsup$  \\
\object{Sz~49}				& 13:00:53.26 & $-$76:54:14.98 & 14.86~$\pm$~0.01	      & 13.53~$\pm$~0.03       & 13.01~$\pm$~0.01 & 13.90~$\pm$~0.09 & 14.93~$\pm$~0.13 & 13.62~$\pm$~0.09 & 13.35~$\pm$~0.08 & a	& $\bigstar$  $\ding{80}$ 	      \\
\object{Sz~48SW}			& 13:00:53.56 & $-$77:09:08.28 & 16.26~$\pm$~0.01	      & 14.50~$\pm$~0.02       & 13.72~$\pm$~0.01 & 16.13~$\pm$~0.09 & 16.57~$\pm$~0.13 & 14.51~$\pm$~0.09 & 14.19~$\pm$~0.08 & a	& $\bigstar$ $\ding{68}$$\leftthumbsup$   \\
\object{Sz~50}				& 13:00:55.28 & $-$77:10:22.01 & 14.42~$\pm$~0.01	      & 12.50~$\pm$~0.01$^\ast$& 11.96~$\pm$~0.02 & 14.34~$\pm$~0.10 & 14.74~$\pm$~0.13 & 12.79~$\pm$~0.09 & 12.43~$\pm$~0.08 & a	& $\bigstar$ $\leftthumbsup$	      \\
\object{WFI~J13005531-7708295}		& 13:00:55.31 & $-$77:08:29.54 & 14.93~$\pm$~0.01	      & 13.29~$\pm$~0.01	     & 12.64~$\pm$~0.01 & 15.26~$\pm$~0.14 & 15.24~$\pm$~0.13 & 13.39~$\pm$~0.09 & 13.15~$\pm$~0.09 & m   & $\leftthumbsup$$^{NEW}$		    \\
\object{RXJ1301.0-7654a}		& 13:00:56.22 & $-$76:54:01.76 & 11.59~$\pm$~0.01$^\ast$& 10.54~$\pm$~0.01$^\ast$& 10.17~$\pm$~0.01 & 11.63~$\pm$~0.09 & 11.71~$\pm$~0.14 & 10.61~$\pm$~0.09 & 10.66~$\pm$~0.09 & d   & $\bigstar$$\leftthumbsup$		 \\
\object{IRAS~F12571-7657}		& 13:00:59.21 & $-$77:14:02.80 & 18.62~$\pm$~0.01	      & 16.16~$\pm$~0.01	     & 15.00~$\pm$~0.01 & 19.26~$\pm$~0.14 & 19.23~$\pm$~0.16 & 16.48~$\pm$~0.23 & 15.79~$\pm$~0.14 & e, g & $\bullet$ $\leftthumbsup$		 \\
\object{ISO-CHA~II~73}		& 13:01:46.03 & $-$77:16:02.89 & 16.31~$\pm$~0.02	      & 15.12~$\pm$~0.01	     & 14.68~$\pm$~0.01 & 16.71~$\pm$~0.22 & 16.55~$\pm$~0.13 & 15.38~$\pm$~0.09 & 15.13~$\pm$~0.08 & g   & $\bullet$ $\leftthumbsdown$  	 \\
\object{WFI~J13014752-7631023}		& 13:01:47.52 & $-$76:31:02.32 & 16.83~$\pm$~0.05       & 15.23~$\pm$~0.01	     & 14.65~$\pm$~0.06 & 16.84~$\pm$~0.20 & 17.05~$\pm$~0.21 & 15.77~$\pm$~0.17 & 15.18~$\pm$~0.12 & m   & $\leftthumbsup$$^{NEW}$		    \\
\object{Sz~51}				& 13:01:58.94 & $-$77:51:21.74 & 13.47~$\pm$~0.02$^\ast$& 12.38~$\pm$~0.01$^\ast$& 11.80~$\pm$~0.02 & 12.46~$\pm$~0.09 & 13.37~$\pm$~0.13 & 12.55~$\pm$~0.09 & 12.29~$\pm$~0.08 & a   & $\bigstar$  		       \\
\object{CM~Cha}				& 13:02:13.49 & $-$76:37:57.68 & 12.93~$\pm$~0.01       & 11.84~$\pm$~0.01       & 11.18~$\pm$~0.01 & 12.15~$\pm$~0.10 & 12.57~$\pm$~0.13 & 11.55~$\pm$~0.09 & 11.62~$\pm$~0.08 & b   & $\bigstar$ $\leftthumbsup$	       \\
\object{C~50}				& 13:02:22.82 & $-$77:34:49.51 & 17.78~$\pm$~0.01       & 15.49~$\pm$~0.02       & 14.53~$\pm$~0.01 & 17.78~$\pm$~0.10 & 18.19~$\pm$~0.20 & 15.48~$\pm$~0.09 & 14.93~$\pm$~0.08 & f, g & $\bullet$ $\leftthumbsup$	       \\
\object{IRAS~12589-7646}		& 13:02:47.73 & $-$77:02:46.32 &   S		      &	     S  	     &	S	      & 12.72~$\pm$~0.09 & 12.92~$\pm$~0.13 &   S	         &   S		  & i   & $\bullet$			    \\
\object{RXJ1303.1-7706}			& 13:03:04.46 & $-$77:07:02.75 & 12.27~$\pm$~0.01$^\ast$& 11.17~$\pm$~0.01$^\ast$& 10.80~$\pm$~0.01 & 12.34~$\pm$~0.09 & 12.08~$\pm$~0.13 & 11.29~$\pm$~0.09 & 11.16~$\pm$~0.08 & c   & $\bigstar$ $\leftthumbsup$	    \\
\object{C~51}				& 13:03:09.04 & $-$77:55:59.52 & 16.88~$\pm$~0.01	      & 14.47~$\pm$~0.01	     & 13.46~$\pm$~0.02 & 16.82~$\pm$~0.10 & 17.16~$\pm$~0.13 & 14.39~$\pm$~0.09 & 14.03~$\pm$~0.10 & f   & $\bullet$ $\leftthumbsup$		  \\
\object{ChaII~376}			& 13:03:12.45 & $-$76:50:50.82 & 16.31~$\pm$~0.01	      & 15.45~$\pm$~0.01	     & 15.35~$\pm$~0.05 & 16.40~$\pm$~0.10 & 16.42~$\pm$~0.13 &	   NO	 &      NO        & j   & $\bullet$ $\leftthumbsdown$ 	  \\
\object{WFI~J13031615-7629381}		& 13:03:16.15 & $-$76:29:38.15 & 16.42~$\pm$~0.01       & 14.41~$\pm$~0.03	     & 13.68~$\pm$~0.03 & 16.30~$\pm$~0.10 & 16.67~$\pm$~0.15 &      NO        &      NO	  & m   & $\leftthumbsup$$^{NEW}$		    \\
\object{ISO-CHA~II~98a}		& 13:03:25.85 & $-$77:01:48.36 & 17.09~$\pm$~0.02       & 16.42~$\pm$~0.04       & 16.24~$\pm$~0.03 & 17.53~$\pm$~0.10 & 17.27~$\pm$~0.11 & 16.83~$\pm$~0.09 & 17.17~$\pm$~0.24 & g	& $\bullet$ $\ding{68}$ $\leftthumbsdown$ \\
\object{ISO-CHA~II~98b}		& 13:03:26.10 & $-$77:01:48.40 & 17.09~$\pm$~0.02       & 16.37~$\pm$~0.01       & 16.18~$\pm$~0.02 & 17.47~$\pm$~0.10 & 17.24~$\pm$~0.11 & 17.68~$\pm$~0.43 & 16.99~$\pm$~0.22 & g	& $\bullet$ $\ding{68}$ $\leftthumbsdown$ \\
\object{ISO-CHA~II~110}		& 13:04:18.96 & $-$76:53:59.96 & 14.52~$\pm$~0.01       & 14.03~$\pm$~0.01       & 14.03~$\pm$~0.01 & 14.59~$\pm$~0.09 & 14.58~$\pm$~0.13 &      NO	 &	NO	  & g	& $\bullet$ $\leftthumbsdown$	      \\
\object{Hn~22}  			& 13:04:22.78 & $-$76:50:05.86 & 13.51~$\pm$~0.01$^\ast$& 12.68~$\pm$~0.01$^\ast$& 12.17~$\pm$~0.02 & 13.23~$\pm$~0.09 & 13.85~$\pm$~0.13 &    NO 	 &      NO	  & b   & $\bigstar$ $\ding{68}$		     \\
\object{Hn~23}  			& 13:04:24.06 & $-$76:50:01.39 & 12.23~$\pm$~0.01$^\ast$& 11.33~$\pm$~0.08$^\ast$&     S	      & 12.11~$\pm$~0.09 & 12.34~$\pm$~0.13 &    NO 	 &      NO	  & b   & $\bigstar$ $\ding{68}$ $\leftthumbsup$  \\
\object{Sz~52}  			& 13:04:24.84 & $-$77:52:30.11 & 16.89~$\pm$~0.01       & 14.94~$\pm$~0.01	     & 14.12~$\pm$~0.01 & 17.00~$\pm$~0.10 & 17.35~$\pm$~0.13 & 14.84~$\pm$~0.09 & 14.43~$\pm$~0.08 & a   & $\bigstar$ $\leftthumbsup$	     \\
\object{Hn~24}  			& 13:04:55.74 & $-$77:39:49.21 & 13.30~$\pm$~0.06$^\ast$& 11.95~$\pm$~0.02$^\ast$& 11.50~$\pm$~0.02 & 13.49~$\pm$~0.09 & 13.43~$\pm$~0.14 & 12.19~$\pm$~0.09 & 12.01~$\pm$~0.08 & b	& $\bigstar$ $\leftthumbsup$	     \\
\hline
\object{Hn~25}  			& 13:05:08.51 & $-$77:33:42.66 & 16.11~$\pm$~0.01       & 14.17~$\pm$~0.02	     & 13.36~$\pm$~0.01 & 16.45~$\pm$~0.12 & 16.61~$\pm$~0.13 & 14.40~$\pm$~0.09 & 14.01~$\pm$~0.09 & b   & $\bigstar$ $\leftthumbsup$	     \\
\object{Sz~53}  			& 13:05:12.66 & $-$77:30:52.56 & 15.58~$\pm$~0.02       & 13.86~$\pm$~0.02	     & 13.15~$\pm$~0.01 & 15.58~$\pm$~0.09 & 15.90~$\pm$~0.18 & 15.16~$\pm$~0.12 & 14.47~$\pm$~0.08 & a   & $\bigstar$ $\leftthumbsup$	     \\
\object{Sz~54}  			& 13:05:20.80 & $-$77:39:01.48 & 11.58~$\pm$~0.02$^\ast$& 10.61~$\pm$~0.06$^\ast$& 10.47~$\pm$~0.43 & 11.28~$\pm$~0.10 & 11.57~$\pm$~0.13 & 10.80~$\pm$~0.09 & 11.05~$\pm$~0.15 & a	& $\bigstar$ $\leftthumbsup$	     \\
\object{Sz~55}  			& 13:06:30.49 & $-$77:34:00.12 & 16.37~$\pm$~0.01       & 14.70~$\pm$~0.06	     & 14.06~$\pm$~0.47 & 15.65~$\pm$~0.09 & 16.96~$\pm$~0.13 & 14.61~$\pm$~0.09 & 14.53~$\pm$~0.08 & a   & $\bigstar$ $\leftthumbsup$	     \\
\object{Sz~56}				& 13:06:38.70 & $-$77:30:35.39 & 15.41~$\pm$~0.01$^\ast$& 13.47~$\pm$~0.02$^\ast$& 12.71~$\pm$~0.44 & 15.41~$\pm$~0.09 & 15.75~$\pm$~0.13 & 13.67~$\pm$~0.09 & 13.15~$\pm$~0.08 & a   & $\bigstar$ $\leftthumbsup$	   \\
\object{Sz~57}				& 13:06:56.56 & $-$77:23:09.46 & 15.97~$\pm$~0.01	      & 13.65~$\pm$~0.01	     & 12.59~$\pm$~0.41 & 15.71~$\pm$~0.09 & 16.23~$\pm$~0.13 & 13.74~$\pm$~0.09 & 13.07~$\pm$~0.08 & a   & $\bigstar$ $\leftthumbsup$ 	   \\
\object{Sz~58}				& 13:06:57.35 & $-$77:23:41.46 & 14.63~$\pm$~0.01	      & 13.16~$\pm$~0.01	     & 12.53~$\pm$~0.43 & 14.44~$\pm$~0.09 & 14.58~$\pm$~0.13 & 12.98~$\pm$~0.09 & 12.78~$\pm$~0.08 & a   & $\bigstar$ $\leftthumbsup$ 	   \\
\object{Sz~59}				& 13:07:09.23 & $-$77:30:30.24 & 13.42~$\pm$~0.02$^\ast$& 12.08~$\pm$~0.01$^\ast$& 11.51~$\pm$~0.43 & 12.96~$\pm$~0.10 & 13.39~$\pm$~0.13 & 12.57~$\pm$~0.09 & 12.50~$\pm$~0.08 & a   & $\bigstar$ $\leftthumbsup$	   \\
\object{ C~62}				& 13:07:18.04 & $-$77:40:53.00 & 18.60~$\pm$~0.01	      & 16.10~$\pm$~0.01	     & 15.00~$\pm$~0.44 & 18.42~$\pm$~0.11 & 18.97~$\pm$~0.17 & 16.23~$\pm$~0.09 & 15.60~$\pm$~0.08 & f, l & $\bullet$ $\leftthumbsup$  	   \\
\object{WFI~J13071960-7655476}		& 13:07:19.60 & $-$76:55:47.64 & 18.33~$\pm$~0.01       & 16.49~$\pm$~0.02	     & 15.77~$\pm$~0.02 & 18.12~$\pm$~0.10 & 18.57~$\pm$~0.14 &       NO       &	NO	  & m   & $\leftthumbsup$$^{NEW}$		    \\
\object{Sz~60W}				& 13:07:22.30 & $-$77:37:22.62 & 14.88~$\pm$~0.02$^\ast$& 13.45~$\pm$~0.02$^\ast$& 13.02~$\pm$~0.46 & 14.60~$\pm$~0.09 & 15.01~$\pm$~0.13 & 13.42~$\pm$~0.09 & 13.23~$\pm$~0.07 & a   & $\bigstar$ $\ding{68}$ $\leftthumbsup$  \\
\object{Sz~60E}				& 13:07:23.33 & $-$77:37:23.20 & 15.32~$\pm$~0.02$^\ast$& 13.60~$\pm$~0.01$^\ast$& 12.94~$\pm$~0.46 & 15.01~$\pm$~0.09 & 15.69~$\pm$~0.13 & 13.98~$\pm$~0.09 & 13.61~$\pm$~0.08 & a   & $\bigstar$ $\ding{68}$		    \\
\object{Hn~26}				& 13:07:48.50 & $-$77:41:21.73 & 16.07~$\pm$~0.01	      & 14.31~$\pm$~0.01	     & 13.54~$\pm$~0.46 & 15.89~$\pm$~0.11 & 16.25~$\pm$~0.13 & 14.41~$\pm$~0.09 & 14.13~$\pm$~0.08 & b   & $\bigstar$ $\leftthumbsup$ 	    \\
\object{Sz~61}				& 13:08:06.33 & $-$77:55:05.05 & 13.69~$\pm$~0.02$^\ast$& 12.38~$\pm$~0.01$^\ast$& 11.48~$\pm$~0.42 & 12.69~$\pm$~0.09 & 13.47~$\pm$~0.13 & 12.27~$\pm$~0.09 & 12.14~$\pm$~0.08 & a   & $\bigstar$ $\leftthumbsup$	    \\
\object{C~66}				& 13:08:27.19 & $-$77:43:23.41 & 18.91~$\pm$~0.01       & 16.51~$\pm$~0.01	     & 15.44~$\pm$~0.44 & 19.24~$\pm$~0.14 & 19.72~$\pm$~0.22 & 16.76~$\pm$~0.09 & 16.03~$\pm$~0.08 & f, l & $\bullet$ $\leftthumbsup$ 	      \\
\object{IRAS~F13052-7653NW}		& 13:09:09.81 & $-$77:09:43.52 & 15.37~$\pm$~0.01	      & 14.04~$\pm$~0.02	     & 13.52~$\pm$~0.44 & 15.59~$\pm$~0.09 & 16.11~$\pm$~0.13 & 14.02~$\pm$~0.09 & 13.84~$\pm$~0.08 & e   & $\bullet$ $\ding{68}$		  \\
\object{IRAS~F13052-7653S}		& 13:09:10.67 & $-$77:09:46.84 & 13.49~$\pm$~0.01       & 12.57~$\pm$~0.01       & 12.38~$\pm$~0.47 & 13.09~$\pm$~0.09 & 13.04~$\pm$~0.13 & 12.69~$\pm$~0.09 & 12.77~$\pm$~0.08 & e   & $\bullet$ $\ding{68}$  	       \\
\object{IRAS~F13052-7653N}		& 13:09:10.98 & $-$77:09:44.14 & 13.47~$\pm$~0.01       & 12.42~$\pm$~0.02       & 12.07~$\pm$~0.45 & 13.51~$\pm$~0.09 & 13.74~$\pm$~0.13 & 12.54~$\pm$~0.09 & 12.47~$\pm$~0.08 & e   & $\bullet$ $\ding{68}$  	       \\
\object{Sz~62}				& 13:09:50.44 & $-$77:57:23.94 & 14.03~$\pm$~0.02$^\ast$& 12.56~$\pm$~0.02$^\ast$& 11.66~$\pm$~0.43 &	   NO	       &	  NO    &     NO	 &      NO        & a   & $\bigstar$ $\leftthumbsup$ 	   \\
\object{Sz~63}				& 13:10:04.12 & $-$77:10:44.62 & 14.66~$\pm$~0.01	      & 13.21~$\pm$~0.01       & 12.66~$\pm$~0.44 & 14.40~$\pm$~0.09 & 15.09~$\pm$~0.13 & 13.25~$\pm$~0.09 & 13.03~$\pm$~0.08 & a   & $\bigstar$ $\leftthumbsup$	       \\
\object{2MASS13102531-7729082}    	& 13:10:25.26 & $-$77:29:08.70 & 16.49~$\pm$~0.01	      & 15.36~$\pm$~0.05	     & 14.95~$\pm$~0.48 & 16.49~$\pm$~0.10 & 16.49~$\pm$~0.13 & 15.54~$\pm$~0.09 & 15.43~$\pm$~0.08 & i   & $\bullet$ $\leftthumbsdown$	   \\
\object{2MASS13110329-7653330}    	& 13:11:03.27 & $-$76:53:32.89 & 14.50~$\pm$~0.01	      & 13.85~$\pm$~0.00	     & 13.68~$\pm$~0.01 & 14.62~$\pm$~0.09 & 14.57~$\pm$~0.13 &      NO        &      NO        & i   & $\bullet$ $\leftthumbsdown$	   \\
\object{2MASS13125238-7739182}    	& 13:12:52.37 & $-$77:39:18.58 & 15.36~$\pm$~0.01	      & 13.62~$\pm$~0.02	     & 12.97~$\pm$~0.47 & 15.57~$\pm$~0.09 & 15.78~$\pm$~0.13 & 13.83~$\pm$~0.09 & 13.51~$\pm$~0.08 & i   & $\bullet$ $\leftthumbsup$  	   \\
\object{Sz~64}                    	& 13:14:03.83 & $-$77:53:07.48 & 15.82~$\pm$~0.01       & 14.01~$\pm$~0.01       & 13.16~$\pm$~0.45 & 15.25~$\pm$~0.09 & 16.29~$\pm$~0.13 & 14.40~$\pm$~0.09 & 13.65~$\pm$~0.08 & a   & $\bigstar$ $\leftthumbsup$	       \\
\hline
\end{tabular}
\par
\smallskip
{Notes: The positions for most objects are from the $R$-band images. 
For saturated objects in the $R$-band the positions are from the H$\alpha_{12}$~images; 
the coordinates of \object{Sz~62}, not observed in H$\alpha$, are from the 
$z$-band image. Magnitudes marked with an asterisk are taken from the the literature (see the relative references) 
and correspond to objects saturated in our images. Comments about some of these objects are given in Appendix~\ref{comm}. 
Symbols and labels are as follows:} 
{ 
\begin{itemize}
\item $\bigstar$ = Confirmed member 
\item $\bullet$ = Previously known candidate
\item $\leftthumbsup$  = Object recovered by our selection criteria
\item $\leftthumbsup$$^{NEW}$  = Member candidate firstly identified in this work
\item $\leftthumbsdown$  = Object rejected on the basis of our selection criteria
\item $\ding{80}$ = Veiled object
\item $\ding{74}$ = Embedded object
\item $\ding{68}$ = Component of a visual binary or multiple system
\item S = Saturated object
\item ND = Object not detected 
\item NO = Object not observed 
\end{itemize}
}
\par
{Main references: a  \citet{Hug92}; b  \citet{Har93}; c  \citet{Alc95}; d  \citet{Cov97}; e \citet{Alc00}; f  \citet{Vuo01}; 
g  \citet{Per03}; h  \citet{Bar04}; i  \citet{You05}; j  \citet{Lop05}; k  \citet{Alc06}; l  \citet{All06}; m  This work.}
\par
\end{table}